Author’s response to reviews

Title: Community-based educational design for undergraduate medical education: A grounded theory study

Authors:

Mora Claramita (mora.claramita@ugm.ac.id)
Elsa Setiawati (elsapudji@gmail.com)
Ova Emilia (ovaemilia@gmail.com)
Tri Kristina (t_nurkristina@yahoo.com)
Cees van der Vleuten (c.vandervleuten@maastrichtuniversity.nl)

Version: 2 Date: 23 May 2019

Author’s response to reviews:

Dear the Editorial Office

BMC Medical Education

https://bmcmededuc.biomedcentral.com/

Thank you very much for your comments and feedback, herewith my responses below.

Sincerely,

Mora Claramita (Corresponding author)

FORMATTING CHANGES:

1. Please provide the email addresses of all authors in the title page and please make the title on one page.

Answer: I added one title page with email addresses

2. Please provide List of Abbreviations.
Answer: I provided the List of Abbreviations (after the Conclusion and before Declarations)

3. Please rename "Disclosure statement" to "Competing Interests".

Answer: Yes I renamed it accordingly

DECLARATIONS SECTION:

We note that the Declarations section in your manuscript has not been completed and may be missing information. Please read the following information and edit your manuscript accordingly. Failure to adhere to our policies may result in rejection of your manuscript.

Please note that all manuscript submissions to BMC Medical Education must contain a Declarations section which includes a response to each of the mandatory sub-headings listed below. Please refer to the journal's Submission Guidelines for information regarding the criteria for each sub-section (LINK).

Answer: I have read and revised this section based on the Submission Guidelines via these links:

https://bmcmededuc.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

https://www.biomedcentral.com/getpublished/editorial-policies#Ethics

Where a mandatory Declarations sub-heading is not relevant to your study design or article type, please write "Not applicable" in these sections.

Answer: No individual details, images, or video used in this study.

Consent for publication:

Not applicable.
For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared.

Authors who do not wish to share their data must confirm this under this sub-heading and also provide their reasons.

This is a qualitative study, all data contains of narrative explanation from participants. We cannot share the data and must get an individual permission from every participant if we would do that. This study is different with, for example, an epidemiology study with hundreds of participants, which contain lots of statistics methods. Quantitative data can be shared and can be beneficial to other readers. Whereas qualitative data is specific to only in this study. Herewith what we mentioned regarding the data:

Availability of data and materials:

Data of this qualitative study were obtained from interviews (with the participants) and coding processes (by the research assistants) and were in Indonesian Language. Data for validation of the CBE framework were obtained via electronic mail from international experts and were in English. The data could be retrieved on reasonable request.

For further guidance on how to format this section, please refer to BioMed Central's editorial policies page (see links below).

Yes we did.

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors’ Contributions
- Acknowledgements

Further information about our editorial policies can be found at the following links:
Ethical approval and consent:

http://www.biomedcentral.com/about/editorialpolicies#Ethics

Availability of data and materials section:

http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials